Author’s response to reviews

Title: Bilateral spontaneous non-traumatic rupture of the Achilles tendon - a case report.

Authors:

Zubair S Khanzada (zubairshabbir@hotmail.com)
Ulfin Rethnam (ulfinr@yahoo.com)
David Widdowson (David.Widdowson@wales.nhs.uk)
Ahmad H Mirza (Ahmad.Mirza@wales.nhs.uk)

Version: 2 Date: 6 February 2010

Author’s response to reviews:

1. An abstract has been included on page 2 of the manuscript.
2. Ethnic origin and nationality of the patient has been included in the case presentation section of the manuscript.
3. Authors email addresses have been included on page 1.
4. Competing interests have included on page 6.